Author's response to reviews

Title: Colectomy rate in steroid-refractory colitis initially responsive to cyclosporin: A long-term retrospective cohort study

Authors:

Giovanni C Actis (actis_g@libero.it)
Marco Pinna-Pintor (mpinnapintor@molinette.piemonte.it)
Marco Astegiano (mastegiano@molinette.piemonte.it)
Maurizio Fadda (mfadda@molinette.piemonte.it)
Ezio David (edavid@molinette.piemonte.it)
Mario Rizzetto (mrizzetto@molinette.piemonte.it)
Anna Sapino (asapino@molinette.piemonte.it)

Version: 3 Date: 12 February 2007

Author's response to reviews: see over
Dear Editor:

As suggested in your comment, we expanded the background section of the abstract to cover the purpose and context of the study; in addition, all of the nine changes required by the referee were made; the paragraph which duplicated the data at the beginning of the discussion and table 4 have both been deleted.
The paper was finally revised by Biomedes.
We hope that the current edition of this paper fits your requirements.
Looking forward to hearing from you, we remain

Yours Sincerely

GC Actis, MD
Corresponding Author